

## Please cite the Published Version

Junejo, Rehan T , Gupta, Dhiraj, Snowdon, Richard L, Lip, Gregory YH and Fisher, James P (2024) Relationship of Warfarin and Apixaban with Vascular Function in Patients with Atrial Fibrillation. Journal of Vascular Research, 61 (2). pp. 59-67. ISSN 1018-1172

# DOI: https://doi.org/10.1159/000535618

Publisher: S. Karger AG

Version: Published Version

Downloaded from: https://e-space.mmu.ac.uk/634041/

Usage rights: Creative Commons: Attribution 4.0

**Additional Information:** This is an open access article which appeared in Journal of Vascular Research

**Data Access Statement:** Data which support the findings of this study are not publicly available due to funder restrictions but are available from the corresponding author, upon reasonable request.

## Enquiries:

If you have questions about this document, contact openresearch@mmu.ac.uk. Please include the URL of the record in e-space. If you believe that your, or a third party's rights have been compromised through this document please see our Take Down policy (available from https://www.mmu.ac.uk/library/using-the-library/policies-and-guidelines) **JVR** Journal of Vascular Research

# **Research Article**

J Vasc Res DOI: 10.1159/000535618 Received: April 12, 2023 Accepted: November 30, 2023 Published online: March 6, 2024

# Relationship of Warfarin and Apixaban with Vascular Function in Patients with Atrial Fibrillation

Rehan T. Junejo<sup>a, b</sup> Dhiraj Gupta<sup>b, c</sup> Richard L. Snowdon<sup>c</sup> Gregory Y.H. Lip<sup>b, c</sup> James P. Fisher<sup>d</sup>

<sup>a</sup>Department of Life Sciences, Faculty of Science and Engineering, Manchester Metropolitan University, Manchester, UK; <sup>b</sup>Liverpool Centre for Cardiovascular Science, University of Liverpool, Liverpool, UK; <sup>c</sup>Department of Cardiology, Liverpool Heart and Chest Hospital, Liverpool, UK; <sup>d</sup>Department of Physiology, Faculty of Medical and Health Sciences, Manaaki Manawa – The Centre for Heart Research, University of Auckland, Auckland, New Zealand

#### **Keywords**

Atrial fibrillation  $\cdot$  Flow-mediated dilatation  $\cdot$  Shear stress  $\cdot$  Warfarin  $\cdot$  Apixaban

## Abstract

Introduction: Atrial fibrillation (AF) is associated with endothelial damage/dysfunction. Herein, we tested the hypothesis that brachial artery flow-mediated dilation (FMD) is superior in AF patients taking apixaban compared to warfarin. *Methods:* AF patients on apixaban (n = 46; 67 [7] years; mean [standard deviation]; 15 women) and warfarin (n = 27; 73 [9] years (p < 0.01); 11 women) were recruited. Duplex Doppler ultrasound imaging was undertaken during baseline (2 min), cuff inflation (5 min), and following cuff deflation (3 min). FMD was defined as peak increase in brachial artery diameter following cuff deflation and analysed as percentage change in diameter, as a ratio of FMD, shear rate area under the curve (SR<sub>AUC</sub>; FMD-to-SR<sub>AUC</sub>), and using SR<sub>AUC</sub> as a covariate (FMD<sub>SR</sub>). *Results:* Baseline artery diameter (4.96 [1.14] vs. 4.89 [0.88] mm), peak diameter (5.12 [1.17] vs. 5.14 [0.93] mm), and FMD<sub>SR</sub> (3.89 [3.62] vs. 4.80 [3.60] %) were not different between warfarin and apixaban (p > 0.05;

karger@karger.com www.karger.com/jvr © 2024 The Author(s). Published by S. Karger AG, Basel

This article is licensed under the Creative Commons Attribution 4.0 International License (CC BY) (http://www.karger.com/Services/ OpenAccessLicense). Usage, derivative works and distribution are permitted provided that proper credit is given to the author and the original publisher. analysis of covariance with age, CHA<sub>2</sub>DS<sub>2</sub>-VASc, years since AF diagnosis, number of diabetics, alcohol drinkers, and units of alcohol consumed per week as covariates). Stepwise multiple regression identified independent association of fibrillation, hypertension, and increased age with FMD. **Conclusion:** AF patients on warfarin and apixaban exhibit similar endothelium-dependent vasodilation. Increased blood pressure negatively impacts vasodilator capacity in AF patients. © 2024 The Author(s).

Published by S. Karger AG, Basel

#### Plain Language Summary

Atrial fibrillation (AF) is the most common sustained heart rhythm abnormality and is associated with blood vessel damage/dysfunction. AF patients also have a substantially increased risk of stroke from blood clots. As such, these

Gregory Y.H. Lip and James P. Fisher should be considered as joint senior authors.

The authors confirm that the Principal Investigators for this paper are Prof. Gregory Y.H. Lip and Dr. James P. Fisher. Prof. Gregory Lip had direct clinical responsibility for patients.

Correspondence to: James P. Fisher, jp.fisher@auckland.ac.nz

 patients are prescribed anti-coagulants, such as warfarin and apixaban. Herein, we compared the vascular function of the brachial artery in AF patients taking apixaban to those taking warfarin. Brachial artery's vasodilator capacity was not different between groups, indicating comparable vascular function of AF patients on warfarin and apixaban. Increased resting blood pressure in AF patients negatively impacted dilation capacity of the brachial artery in AF patients.

> © 2024 The Author(s). Published by S. Karger AG, Basel

#### Introduction

Atrial fibrillation (AF) is a major risk factor for severe thromboembolic events, and the irregular cardiac rhythm and hypercoagulable states observed in AF are related to endothelial dysfunction [1]. Indeed, our group [2, 3] and others [4] have reported impaired brachial artery flowmediated dilation (FMD), a marker of endothelial dysfunction, in AF. This is important because poor FMD is associated with adverse cardiovascular events [5], but strategies to enhance endothelial function in AF are limited.

There is a considerable interplay between coagulation, inflammation, and endothelial dysfunction. Endotoxins, C-reactive protein (CRP), and tumour necrosis factor a (TNFa) all promote tissue factor formation, and exposure of tissue factor to factor VIIa catalyses conversion of factor X to Xa, initiating thrombin formation and platelet activation [6]. In a self-perpetuating feedback cycle, coagulation proteases, such as factor Xa and thrombin, then contribute to the upregulation of the inflammatory response by increasing TNFa, interleukin-6 (IL-6), and IL-8 concentrations [6]. Various inflammatory molecules (e.g., CRP, TNFa, IL-2, IL-6, and IL-8) have been linked to the presence and outcome of AF, promoting endothelial damage/dysfunction [7]. Although positive linear correlations between plasma biomarkers of vascular dysfunction (e.g., von Willebrand factor) and coagulation factors (e.g., fibrinogen) have been reported in AF [8], whether oral anti-coagulation improves endothelial function remains unclear.

Apixaban is a direct oral anti-coagulant (DOAC) that inhibits factor Xa, thus decreasing thrombin generation and thrombus development. In the ARISTOTLE trial, apixaban was shown to be superior to warfarin, which non-specifically inhibits all vitamin K-dependent clotting factors, in preventing stroke or systemic embolism [9]. Further, apixaban treatment resulted in lower mortality in patients with AF and reduced incidence of major bleeding [9]. Indeed, apixaban has independently been shown to increase vasodilation and attenuate phenylephrine and 5-hvdroxytryptamine-mediated vasoconstriction [10]. Moreover, in in vitro models of endothelial dysfunction resulting from uraemic toxins, apixaban has been shown to provide endothelial/vascular protection by decreasing ICAM-1, VCAM-1, and von Willebrand factor expression and upregulating expression of endothelial nitric oxide synthase (eNOS) [11]. In light of the emerging evidence for the malign trifecta of coagulation, inflammation, and endothelial dysfunction, we hypothesised that some of this benefit may be ascribable to superior FMD in patients taking apixaban compared to warfarin. This study was thus designed to investigate the influence of these two oral anti-coagulants on brachial artery FMD in AF.

## Methods

#### Ethical Approval

All procedures were undertaken in accordance with the Declaration of Helsinki and were approved by the National Research Ethics Service Committee North West (17/NW/0714). Research funding support was provided by Bristol-Myers Squibb-Pfizer Alliance. Prospective participants were provided with a detailed verbal explanation of study procedures and an information sheet. Written informed consent was obtained from all participants.

#### Participant Characteristics

Seventy-three AF patients from predominantly white European ethnic background were recruited across 2 anti-coagulation groups based on their current anti-coagulant use: warfarin (n = 27) and apixaban (n = 46; Table 1). Patients were previously clinically diagnosed with at least paroxysmal AF (i.e., episodes are transient and spontaneously resolve within 48 h) and were recruited from dedicated cardiology clinics at City Hospital, Birmingham, at Liverpool Heart and Chest Hospital, Liverpool, and from West Midlands GP practices within the National Institute of Health Research Clinical Research Network. AF diagnosis was confirmed in nearly all patients at least a year prior to study recruitment, except 2 apixaban patients. Premenopausal women, individuals with valvular heart disease, left ventricular dysfunction, myocardial infarction, respiratory, hepatic, renal, inflammatory, connective tissue, neurological, malignant diseases, stroke, transient ischaemic attack (<3 years), uncontrolled thyroid disorders, and current smokers were excluded from participation. Prescribed and over-the-counter supplemental medications are listed in Table 2.

#### **Experimental Measures**

Medical history, medication use, smoking history, and current alcohol use were recorded. CHA<sub>2</sub>DS<sub>2</sub>-VASc scores were calculated using the medical history. Anthropometric measures of height, weight, waist (level of umbilicus), and hip (level of femoral trochanter) circumference were obtained. Brachial blood pressure (BP) and heart rate (HR) were determined non-invasively using an automated oscillometric device (M2,

		C C	/
and wall tracking softw	are (Cardiova	iscular Suite Ve	ersion 3.4.1,
FMD Studio, Pisa, Italy	). Vascular m	easurements w	ere made in

#### Experimental Protocol

described in detail below.

Patients were requested to abstain from vigorous exercise, food, and caffeine for 12 h and alcohol for 24 h prior to their experimental sessions. Further, patients refrained from taking their usual morning medications on the study day, except for anti-coagulants. Participants were asked to lie supine on a medical examination couch with their head on a pillow, while all assessments took place. Three resting measures of right brachial BP and HR were taken, separated by ~ 2 min and averaged. A

Omron, Kyoto, Japan). Duplex Doppler ultrasound (GE LOGIQ

e, GE Healthcare China Corporation, Jiangsu, China) was used

to simultaneously obtain right brachial artery diameter and flow

velocity measurements while the arm was supported at heart

level. The artery was insonated at  $60^{\circ} \sim 10-15$  cm proximal to the medial epicondyle. A 5-13 MHz multi-frequency linear-

array transducer (GE 12L-RS, GE Healthcare Japan Corporation, Tokyo, Japan) was used for B-mode imaging of vessel diameter and the pulse wave mode used for assessment of peak

blood velocity. Recordings were screen captured and stored as

video files for offline analysis with automated edge detection

accordance with the recent technical recommendations [12], as

Atrial Fibrillation and Vascular Function

#### Table 1. Participant characteristics

	Warfarin ( $n = 27$ )	Apixaban ( $n = 46$ )	$p$ value ( $d$ , $\phi$ )
Age, years Female, $n$ (%) Weight, kg Height, m BMI, kg m <sup>-2</sup> Waist, cm Hip, cm Waist-to-hip Waist-to-height Systolic BP, mm Hg Diastolic BP, mm Hg MBP, mm Hg HR, b min <sup>-1</sup> Years since AF diagnosis Non-fibrillating, $n$ (%) Hypertension, $n$ (%) Diabetes, $n$ (%) Diabetes, $n$ (%) CHA <sub>2</sub> DS <sub>2</sub> -VASc score Previous smoker, $n$ (%) Alcohol users, $n$ (%)	73 $\pm$ 9 11 (41) 88 $\pm$ 20 1.72 $\pm$ 0.09 29 $\pm$ 4 98 $\pm$ 10 109 $\pm$ 10 0.88 [0.85–0.96] 0.57 $\pm$ 0.05 148 $\pm$ 18 84 $\pm$ 10 105 $\pm$ 11 67 [59–70] 10 $\pm$ 8 17 (63) 18 (67) 4 (15) 3 [2–3] 17 (63) 16 (59)	$\begin{array}{c} 67\pm7\\ 15\ (33)\\ 86\pm18\\ 1.72\pm0.09\\ 29\pm5\\ 97\pm12\\ 106\pm9\\ 0.91\ [0.84-0.99]\\ 0.56\pm0.07\\ 144\pm15\\ 85\pm13\\ 105\pm11\\ 69\ [64-74]\\ 5\pm4\\ 24\ (52)\\ 23\ (50)\\ 0\\ 2\ [1-2]\\ 22\ (48)\\ 40\ (87) \end{array}$	0.002 (0.744) 0.484 0.726 0.959 0.846 0.709 0.213 0.827 0.670 0.208 0.681 0.811 0.665 0.009 (0.791) 0.370 0.166 0.016 (0.314) <0.001 (0.921) 0.211 0.026 (0.317)
Alcohol (units), week <sup>–1</sup>	2 [0–8]	7 [2–12]	0.049 (0.450)

Values expressed as the mean  $\pm$  SD for normally distributed data, as median [interquartile range] for nonnormally distributed data, and as frequency (percentage) for discrete variables. Statistical differences were tested using an independent *t* test or Mann-Whitney U test for continuous variables and  $\chi^2$  test for categorical data. If significant, the effect size was reported as Cohen's *d* or Phi for continuous and categorical variables, respectively. Significance: p < 0.05.  $\varphi$ , Phi; BMI, body mass index; *d*, Cohen's *d*; MBP, mean blood pressure.

narrow inflatable tourniquet cuff (Hokanson, Bellevue, WA, USA) was then wrapped around the forearm ~7 cm distal to the medial epicondyle. The FMD protocol comprised a 2-min baseline, a 5-min supra-systolic cuff inflation (>240 mm Hg), and a 3-min recovery period following cuff deflation. Simultaneous measurements of brachial artery diameter and flow velocity were undertaken throughout.

#### Data Analysis

Waist-to-hip and waist-to-height ratios were calculated, along with body mass index (ratio of weight and height squared). Brachial artery measures were calculated as a 2 min average for the baseline period and on a second-by-second basis during the postcuff deflation period. FMD was taken as the maximal change in brachial artery diameter following cuff deflation and expressed as absolute (mm) and relative (%) change. Time-to-peak diameter was obtained between the cuff deflation and the maximal artery dilation. Shear rate (SR) was calculated as follows:

Shear rate = 
$$\frac{\text{Brachial artery flow velocity} \cdot 4}{\text{Brachial artery diameter}}$$

Shear rate area under the curve ( $SR_{AUC}$ ) was calculated as an integral between the cuff deflation and the maximal artery dilation. In accordance with guidelines [12],  $SR_{AUC}$  was used as a covariate to calculate SR-corrected FMD (FMD<sub>SR</sub>) and

	Warfarin ( $n = 27$ )	Apixaban ( $n = 46$ )	p value
Antiplatelets, <i>n</i> (%)	0	2 (4)	0.527
a inhibitor, <i>n</i> (%)	2 (7)	2 (4)	0.623
β inhibitor, <i>n</i> (%)	21 (78)	26 (57)	0.067
Cardiac glycoside, n (%)	2 (7)	10 (22)	0.190
Anti-arrhythmia, n (%)	5 (19)	6 (13)	0.522
ACE inhibitor, n (%)	10 (37)	10 (22)	0.157
Ca <sup>2+</sup> channel inhibitor, <i>n</i> (%)	4 (15)	11 (24)	0.353
ARB, n (%)	6 (22)	7 (15)	0.532
Loop diuretic, <i>n</i> (%)	2 (7)	2 (4)	0.623
Thiazide diuretic, n (%)	1 (4)	0	0.370
Biguanide, n (%)	3 (11)	0	0.047 (0.270)
PP inhibitor, n (%)	10 (37)	11 (24)	0.232
Statin, <i>n</i> (%)	16 (59)	22 (48)	0.345
SSR inhibitor, <i>n</i> (%)	0	4 (9)	0.290
Tricyclic antidepressant, n (%)	2 (7)	1 (2)	0.551
SNR inhibitor, n (%)	0	1 (2)	1.000
NSAID, n (%)	3 (11)	4 (9)	0.705
Folic acid, n (%)	0	1 (2)	1.000

Table 2. Medication use

Values expressed as frequency (percentage) for discrete variables. Statistical differences were tested using  $\chi^2$  test for categorical data. Significance: p < 0.05.  $\alpha$ , alpha;  $\beta$ , beta; ACE, angiotensin converting enzyme, ARB, angiotensin receptor blocker; Ca, calcium; NSAID, non-steroidal anti-inflammatory drug; PP, proton pump; SNR, serotonin noradrenaline reuptake; SSR, selective serotonin reuptake.

reported as estimated margin means. In addition, FMD responses were also normalised for SR by calculating FMD-to-SR<sub>AUC</sub> ratio. Further, in accordance with guidelines for the allometric scaling of baseline artery diameters [13], slope and upper bound 95% confidence intervals of the log-transformed baseline and peak diameters were calculated. As the slope (0.99) and 95% confidence intervals (1.03) were close to unity, allometric scaling was not performed.

Brachial velocity profiles obtained from the Doppler trace are reported as net velocity (positive minus negative velocity profiles). Time-to-peak velocity was obtained between cuff deflation and peak net brachial velocity. Brachial artery blood flow (mL min<sup>-1</sup>) was calculated as follows:

Brachial artery blood flow

$$= \left[\frac{\text{Peak Envelope Velocity}}{2} \cdot (\pi (0.5 \cdot \text{Diameter})^2)\right] \cdot 60$$

Peak hyperaemia was taken as the peak brachial artery blood flow following cuff deflation. Time-to-peak hyperaemia was obtained between cuff deflation and peak blood flow response.

## Statistical Analysis

The distribution of continuous variables was assessed using a Kolmogorov-Smirnov test. Non-normally distributed data were transformed using the natural logarithm method. Data (original and log-transformed) were screened for extreme outliers ( $3 \times$  interquartile range), and any such individual points were excluded from analysis. Normally distributed data were analysed using an independent two-tailed Student's *t* test. Age, years since AF

diagnosis, CHA2DS2-VASc scores, number of diabetics, alcohol drinkers, units of alcohol consumed per week, and biguanide use were identified as being significantly different between groups (Tables 1, 2). Correlation between these variables was assessed using Pearson's r or Spearman's rho, as appropriate, before they were incorporated as covariates in a one-way analysis of covariance (ANCOVA) to control for their influence on further group comparisons/analysis. Correlation between number of diabetics and biguanide use was greater than 0.8; as such, biguanide use was excluded from ANCOVA. Levene's test was used to test equality of error variances between groups. Data that remained non-normally distributed were analysed using a Mann-Whitney U test. Categorical data were analysed using a either a Pearson  $\chi^2$  or Fisher's exact test as appropriate. If significant, Cohen's d(d) and Phi value ( $\varphi$ ) were reported for continuous and categorical variables, respectively. When comparisons were conducted for log-transformed data, transformed values were used for *d* calculations. All continuous and dichotomous variables presented in Tables 1 and 2 (with >5 data points) were entered into stepwise multiple regression to determine their influence on brachial FMD and FMD-to-SRAUC (log-transformed). Multicollinearity and autocorrelation were tested to reduce the risk of regression model violations: reported variables had tolerance of 0.86-0.97, variance inflation factor of 1.03-1.16, and Durbin-Watson statistic of 2.01-2.40.

Data are expressed as mean ± standard deviation (SD) for normally distributed data, median [inter-quartiles] for nonnormally distributed data, and frequency (percentage) for categorical variables, unless stated in legends. Data presented are original and not log-transformed data or ANCOVA estimated

Junejo/Gupta/Snowdon/Lip/Fisher

	Warfarin	Apixaban	p value
Baseline diameter, mm	4.96±1.14	4.89±0.88	0.243
Baseline velocity, cm s <sup>–1</sup>	8.93 (6.18)	9.86 (4.95)	0.876
Baseline blood flow, mL min <sup>-1</sup>	38.09 [28.78–62.86]	51.28 [27.74–76.66]	0.445
Peak diameter, mm	5.12±1.17	5.14±0.93	0.272
lime-to-peak diameter, s	67±31	63±28	0.405
Peak velocity, cm s <sup>-1</sup>	63.01±17.29	75.58±20.04	0.792
lime-to-peak velocity, s	10±4	10±3	0.299
Peak hyperaemia, mL min <sup>-1</sup>	290.28 [227.31–444.12]	412.67 [301.84–542.96]	0.360
lime-to-peak hyperaemia, s	9 [7–11]	11 [8–13]	0.072
$SR_{AUC}$ , s <sup>-1</sup>	11,600±6,073	13,049±6,132	0.634
Absolute FMD, mm	0.16 [0.08–0.24]	0.23 [0.12-0.32]	0.825
-MD, %	3.29±2.99	5.15±3.83	0.715
-MD-to-SR <sub>AUC</sub> , % s <sup>-1</sup>	0.30 [0.16-0.46]	0.39 [0.26–0.63]	0.767
MD <sub>sr</sub> , %	3.89±3.62	4.80±3.60	0.642

Table 3. FMD characteristics in AF patients anti-coagulated with warfarin or apixaban

Values expressed as the mean ± SD for normally distributed variables and as median [interquartile range] for non-normally distributed variables. Statistical differences were tested using one-way ANCOVA with age, CHA2DS2-VASc, years since AF diagnosis, number of diabetics, alcohol drinkers, and units of alcohol consumed per week as covariates; non-normally distributed variables were natural log-transformed before ANCOVA. FMD<sub>SR</sub> ANCOVA was performed with SR<sub>AUC</sub> as an additional covariate; values reported are estimated margin means. Significance: p < 0.05. FMD, flow-mediated dilatation; SR<sub>AUC</sub>, positive shear rate area under the curve.

margin means, unless stated in legends. Correale et al. [14] reported an FMD of 7.5  $\pm$  7.9% and 31.0  $\pm$  15.1% between their AF and heart failure (HF) patients who had been on at least 4 months of either Warfarin or DOAC, respectively. This equates to an effect size of 1.95. Assuming a similar effect size in our AF groups, a sample size of only 8 patients in each group would have determined a statistically significant difference at 95% power and 5% alpha between warfarin and apixaban-treated AF patients. *p* < 0.05 was considered statistically significant. Analysis was performed using IBM SPSS Statistics, version 26 (IBM Corp., Armonk, NY, USA). Data that support the findings of this study are available on reasonable request from the corresponding author. Restrictions apply to availability of these data. Data are not publicly available due to privacy, ethical, and/or 3rd party restrictions.

## Results

## Participant Characteristics

Participant characteristics and medication use are presented in Tables 1 and 2, respectively. Aside from age, there were no significant differences in anthropometric and baseline cardiovascular (e.g., BP and HR) measures. Patients in warfarin group were diagnosed significantly earlier than those on apixaban (10  $\pm$  8 vs. 5  $\pm$  4 years, respectively, p = 0.009). Four diabetics were part of the warfarin group. Therefore, the proportion of diabetics was higher in the warfarin group compared to apixaban

(15% vs. 0%, respectively, p = 0.016), as was the CHA<sub>2</sub>DS<sub>2</sub>-VASc score. Medication use was similar in the warfarin and apixaban groups, aside from biguanide use which was higher in the warfarin group (11% vs. 0%; p =0.047). The proportion of individuals who consumed alcohol and the total units of alcohol consumed per week were greater in apixaban group compared to warfarin (Table 1).

## Brachial Artery FMD

Baseline brachial artery diameter, velocity, and blood flow were not different between the two groups (Table 3; p > 0.05). FMD, whether measured as an absolute, percentage or normalised to SR (FMD-to-SR<sub>AUC</sub> and FMD<sub>SR</sub>), was not different between warfarin and apixaban (Table 3; p > 0.05). Time-to-peak FMD (i.e., peak arterial diameter) and SR<sub>AUC</sub> responses were also not different between the two groups (p > 0.05; Table 3). Peak velocity, peak hyperaemia, and time-to-peak hyperaemia were also similar between groups (p > 0.05; Table 3).

## Stepwise Multivariate Linear Regression

Cardiac arrhythmia (presence of AF at time of investigation), hypertension, and age significantly predicted FMD (F(3,66) = 6.054, p = 0.001), accounting for ~22% of Downloaded from http://karger.com/jvr/article-pdf/doi/10.1159/000535618/4176595/000535618.pdf by Manchester Metropolitan Univ. user on 07 March 2024

**Table 4.** Stepwise multiple regression with FMD (%) and FMD-to-SR<sub>AUC</sub> (%  $s^{-1}$ ) as dependent variables

	β	В	95% Cl for B	p value
Dependent variable: FMD, % $R = 0.465; R^2 = 0.216; adjust$	b = 0.1	80; <i>p</i> = 0.0	01	
Cardiac arrhythmia (AF)	0.315	2.285	0.681-3.889	0.006
Hypertension	-0.360	-2.615	-4.244 to -0.986	0.002
Age, years	-0.232	-0.113	-0.220 to -0.005	0.040
Dependent variable: FMD-to-SR <sub>AUC</sub> (%·s <sup>-1</sup> )				
$k^2 = 0.422; R^2 = 0.178;$ adjusted $R^2 = 0.151; p = 0.003$				
Waist, cm	0.435	0.013	0.005-0.020	0.001
MBP, mm Hg	-0.286	-0.008	-0.015 to -0.001	0.026

Data include warfarin and apixaban groups (n = 73). Variables reported in Tables 1 and 2 were entered in stepwise multiple regression with FMD (%) and FMD-to-SR<sub>AUC</sub> (% s<sup>-1</sup>) as dependent variables; independent variables with  $\leq 5$  data points were excluded from the model. Tolerance for each reported variable was 0.86–0.97, variance inflation factor 1.03–1.16, and Durbin-Watson 2.01–2.40.  $\beta$ , standardised coefficient; B, unstandardised coefficient; CI, confidence interval; *R*, correlation coefficient; *R*<sup>2</sup>, coefficient of determination; MBP, mean blood pressure.

the variation (Table 4). Hypertension and yearly increase in age were negatively correlated with FMD, with a unique contribution of 2.61% and 0.11%, respectively. However, loss of normal cardiac rhythm (i.e., AF) was positively correlated with FMD with a unique contribution of 2.28%. Waist circumference and mean BP significantly predicted FMD-to-SR<sub>AUC</sub> (F(2,61) = 6.607, p = 0.003), accounting for ~18% of the variation. Increase in waist size was positively correlated with FMD-to-SR<sub>AUC</sub> providing a unique contribution of 1.31%.s<sup>-1</sup>, while unit increase in mean BP was negatively correlated with FMD-to-SR<sub>AUC</sub> by 0.80%.s<sup>-1</sup>.

## Discussion

The purpose of the present study was to investigate the hypothesis that the brachial artery FMD is superior in patients taking apixaban compared to warfarin. Contrary to our hypothesis, FMD was not significantly different in AF patients on either apixaban or warfarin. Given the physiological associations between FMD with endothelial/vascular function [15], and with cardiovascular outcomes [5], these findings support the concept that peripheral endothelial/vascular function in AF patients is similar with either apixaban or warfarin oral anti-coagulation.

Exposure of tissue factor to factor VIIa catalyses the conversion of factors IX and X to their enzymatic forms (IXa and Xa), initiating thrombin formation and platelet activation [6]. Evidence implicates inflammation in the

process of thrombus formation and clot generation as well. Indeed, some inflammatory molecules (e.g., endotoxins, CRP, and TNFa) can promote tissue factor formation. Moreover, coagulation proteases (e.g., factor Xa, thrombin) may increase TNFa, IL-6, and IL-8 concentration [6], all of which are associated with the presence and outcome of AF, and promoting endothelial damage/ dysfunction [7]. Mondillo et al. [8] reported that plasma markers of endothelial function, namely, von Willebrand factor and soluble thrombomodulin, were elevated in chronic non-rheumatic AF patients versus controls and were positively correlated with plasma fibrinogen. Moreover, Correale et al. [14] observed that FMD was improved when patients with both AF and chronic HF were switched from Warfarin to a DOAC (e.g., apixaban) for 4 months. This finding may be explained by a superior influence of DOACs on endothelial function [10] and their putative anti-inflammatory effects [16]. However, this could alternatively be attributable to the initial poor patient compliance with warfarin, given that this was the reason that patients were switched to a DOAC in their study. Long-term warfarin use has been associated with vascular calcification and endothelial dysfunction (i.e., increased pulse wave velocity and attenuated finger artery reactive hyperaemia) in AF [17], while changing AF patients from warfarin to the DOAC rivaroxaban for 6 months improves large artery stiffness (i.e., reduces estimated aortic pulse wave velocity and augmentation index) [18].

Herein, we tested the hypothesis that in patients with AF, oral anti-coagulation with apixaban would be associated

with superior FMD than warfarin, but in contrast to expectation, FMD was not different between groups, perhaps on account of the better compliance with warfarin as an anti-coagulant by patients in our study. Indeed, warfarin inhibits multiple enzymatic factors within the intrinsic and extrinsic pathways of coagulation cascade [19], release of which has been linked to proinflammatory molecules [6]. However, we acknowledge that data on compliance with anti-coagulation regimens was not recorded in our patients, and this assumption is speculative; future studies in these study populations should record these data. Additionally, it is also possible that the improvements in FMD with DOAC in Correale et al. [14] are attributable to simultaneous presence of AF and HF, two cardiac conditions which are both independently associated with endothelial dysfunction [2, 20, 21]. Indeed, participants in our study were not diagnosed with HF. Lastly, statins also mitigate the release of inflammatory markers. Indeed, Lenart-Migdalska et al. [22] have noticed a DOAC-related increase in prothrombotic factors is mitigated by statins. Thus, it is possible that our results of similar macrovascular conduit artery endothelial FMD responses between the two anticoagulation groups are in part affected by additional drug regimens (e.g., statins).

Time since AF diagnosis was recorded in our study. However, time since start of anti-coagulation was not obtained. In clinical practice,  $CHA_2DS_2$ - $VAS_C$  criteria help determine need for anti-coagulation, and for most patients, anti-coagulation starts at the time of AF diagnosis. This important consideration was accounted for as a covariate in statistical analysis (i.e., duration of AF diagnosis), and brachial artery FMD was not different between warfarin and apixaban groups.

Endothelial dysfunction is observed in heavy [23] and even in some light alcohol consumers [24]. Given that alcohol use is also associated with AF [25] and both AF and alcohol are linked to endothelial dysfunction, it was possible that our findings of no differences in FMD in warfarin and apixaban groups was affected by significantly greater proportion of alcohol consumers and units of alcohol consumed per week in apixaban group. However, these factors were accounted for in statistical analysis as covariates, and brachial artery FMD was found to be not different between warfarin or apixaban groups.

Previous reports have identified impaired FMD in hypertension [26]. Further, FMD is not different between patients with hypertension versus those with both hypertension and AF [3]. Interestingly, in the current study, regression analysis suggests presence of AF is associated with a better FMD response which is contrary to expectation and previous observations [4]. However, Siasos et al. [27], Komatsu et al. [28], and Khan et al. [3] have shown that the frequency and duration of AF are better associated with attenuated FMD response compared to cardiac rhythm per se. Indeed, Siasos et al. [27] noted that in their cohort, the duration of AF events was the only factor that inversely affected FMD. This may be attributed to mechanosensitive endothelial cells responding to a consistently erratic shear response, leading to proinflammatory gene expression and reduced production of nitric oxide [29]. Conversely, the return to laminar shear is associated with upregulated eNOS gene expression [29]. Therefore, duration and frequency of AF appear to be better determinants of FMD than cardiac rhythm per se. Consistent with this is our observation of a marginally, but not significantly, larger FMD in our cohort who were diagnosed with AF <5 years ago ( $5.09 \pm 4.07$ vs.  $3.74 \pm 2.95$ , respectively; p = 0.11). Age is a wellknown risk factor for endothelial dysfunction [30]. Therefore, poorer FMD in older participants was not an unexpected finding. Lastly, contrary to our expectation, waist size and FMD-to-SR<sub>AUC</sub> were positively correlated. However, whether this reflects use of statins by some participants [31], presence of metabolically healthy obesity [32], or a combination of factors remains to be conclusively investigated. Indeed, Karelis et al. [33] have previously observed lower systemic inflammation, a factor associated with endothelial dysfunction [34, 35], in metabolically healthy individuals compared to insulin-resistant individuals.

## Experimental Considerations

Group size remains uneven between warfarin and apixaban which can affect the statistical assumptions of equal variance between groups. However, significance values for Levene's test of equal variance remained >0.05 throughout between group ANCOVA comparisons which is reassuring for our statistical approach. Nonetheless, future studies comparing endothelial effects of warfarin and DOACs should aim for an evenly matched larger group size. FMD was used to non-invasively assess endothelial/vascular health of patients and guidelines [12, 13] were followed throughout. Future studies however could directly obtain arterial/endothelial tissue from AF patients to investigate the specific signalling pathways for our findings. Blood biomarkers of inflammation and endothelial health were not assessed. Previous studies have observed some differences in circulating biomarkers of vascular health and coagulation [36]. We acknowledge that future work should investigate the influences of anti-coagulants, circulating

Atrial Fibrillation and Vascular Function

blood biomarkers, and vascular function in same cohorts. Endothelium-independent dilatation was not assessed using administration of nitrate as has been previously undertaken in AF [2]. Participants in our group were well matched in terms of proportion of men and women, body mass index, BP, HR, and medication use (aside from biguanide and anti-coagulation). However, age, CHA2DS2-VASc score, years since AF diagnosis, number of diabetics, alcohol drinkers, and units of alcohol consumed per week were different between groups. Given that these might reasonably have been expected to affect FMD, these variables were used as covariates in the between-group assessment of FMD. The relatively stringent inclusion/exclusion criteria we used to limit comorbidities influencing the main outcome variables, the CHA2DS2-VASc score was relatively low in both groups, and this may limit the broader generalisation of our findings to AF patients with significant and common comorbidities (e.g., HF). Although there were only four diabetics in this study, future works should investigate the interplay between independent influences of hyperglycaemia and specific anticoagulants on endothelial function in AF patients. While the brachial artery is commonly used in studies of peripheral vascular function due in part to it providing an accessible representation of the coronary arteries [37], further studies are required to extend our observations to other clinically relevant circulations (e.g., cerebral blood vessels) in which diminished functioning in AF has been identified [38, 39]. All patients were asked to withhold medications on the morning that the study was conducted; however, for clinical reasons, anticoagulants were not withheld. For similar reasons, an AF control group that was anti-coagulation naive was not included. We employed a cross-sectional assessment of non-blinded AF patients taking either apixaban or warfarin, and we acknowledge that a more robust approach would have been to undertake a blinded, longitudinal study of the same patients. However, we believe that the real-world cohorts investigated make our observations representative of the clinic.

## Conclusions

Despite the hypothesis of superiority of apixaban to provide endothelial protection, findings of this study provide important translational evidence that brachial artery FMD of AF patients is not different between those on the vitamin K inhibitor warfarin and the DOAC apixaban, indicative of a similar effects of these oral anticoagulants on endothelial function. Higher BP was associated with decreased brachial artery vasodilator capacity in AF patients.

## Acknowledgment

The authors thank Dr. Alana Cavadino (University of Auckland) and are grateful to her for her expert statistical support.

## **Statement of Ethics**

All procedures were undertaken in accordance with the Declaration of Helsinki and were approved by the National Research Ethics Service Committee North West (17/NW/0714). Prospective participants were provided with a detailed verbal explanation of study procedures and an information sheet. Written informed consent was obtained from all participants.

## **Conflict of Interest Statement**

G.Y.H.L. is a consultant and speaker for BMS/Pfizer, Boehringer Ingelheim and Daiichi-Sankyo. No fees are directly received personally. J.P.F. received funding from BMS/Pfizer for an investigator-led and competitively reviewed research project. R.-T.J. was employed by University of Birmingham as a research fellow to work and manage the BMS/Pfizer funded, investigatorled and competitively reviewed research project. This required regular contact with funders for continued project grant support. No fees/funds were directly received personally.

## **Funding Sources**

Bristol-Myers Squibb-Pfizer Alliance provided the research funding support (investigator-led competitively reviewed research project grant [CV185-481]). The work was also supported by National Institute of Health Research Clinical Research Network (NIHR CRN).

## **Author Contributions**

Gregory Y.H. Lip and James P. Fisher: conception and design; Rehan T. Junejo: data collection, extraction, analysis, figures, and tables; Rehan T. Junejo and James P. Fisher: manuscript draft; Rehan T. Junejo, Richard L. Snowdon, Dhiraj Gupta, Gregory Y.H. Lip, and James P. Fisher: interpretation, revision, critique, and final approval of the manuscript.

#### **Data Availability Statement**

Data which support the findings of this study are not publicly available due to funder restrictions but are available from the corresponding author, upon reasonable request.

## References

- Ding WY, Gupta D, Lip GY. Atrial fibrillation and the prothrombotic state: revisiting Virchow's triad in 2020. Heart. 2020;106(19):1463–8.
- 2 Freestone B, Chong AY, Nuttall S, Lip GY. Impaired flow mediated dilatation as evidence of endothelial dysfunction in chronic atrial fibrillation: relationship to plasma von Willebrand factor and soluble E-selectin levels. Thromb Res. 2008;122(1):85–90.
- 3 Khan AA, Junejo RT, Alsharari R, Thomas GN, Fisher JP, Lip GYH. A greater burden of atrial fibrillation is associated with worse endothelial dysfunction in hypertension. J Hum Hypertens. 2021;35(8):667–77.
- 4 Skalidis EI, Zacharis EA, Tsetis DK, Pagonidis K, Chlouverakis G, Yarmenitis S, et al. Endothelial cell function during atrial fibrillation and after restoration of sinus rhythm. Am J Cardiol. 2007;99(9):1258–62.
- 5 Green DJ, Jones H, Thijssen D, Cable NT, Atkinson G. Flow-mediated dilation and cardiovascular event prediction. Hypertension. 2011;57(3):363-9.
- 6 Levi M, van der Poll T, Büller HR. Bidirectional relation between inflammation and coagulation. Circulation. 2004;109(22):2698–704.
- 7 Guo Y, Lip GY, Apostolakis S. Inflammation in atrial fibrillation. J Am Coll Cardiol. 2012; 60(22):2263–70.
- 8 Mondillo S, Sabatini L, Agricola E, Ammaturo T, Guerrini F, Barbati R, et al. Correlation between left atrial size, prothrombotic state and markers of endothelial dysfunction in patients with lone chronic nonrheumatic atrial fibrillation. Int J Cardiol. 2000;75(2–3):227–32.
- 9 Granger CB, Alexander JH, McMurray JJV, Lopes RD, Hylek EM, Hanna M, et al. Apixaban versus warfarin in patients with atrial fibrillation. N Engl J Med. 2011;365(11):981–92.
- 10 Villari A, Giurdanella G, Bucolo C, Drago F, Salomone S. Apixaban enhances vasodilatation mediated by protease-activated receptor 2 in isolated rat arteries. Front Pharmacol. 2017;8:480.
- 11 Torramade-Moix S, Palomo M, Vera M, Jerez D, Moreno-Castaño AB, Zafar MU, et al. Apixaban downregulates endothelial inflammatory and prothrombotic phenotype in an in vitro model of endothelial dysfunction in uremia. Cardiovasc Drugs Ther. 2021;35(3):521–32.
- 12 Thijssen DH, Bruno RM, van Mil AC, Holder SM, Faita F, Greyling A, et al. Expert consensus and evidence-based recommendations for the assessment of flow-mediated dilation in humans. Eur Heart J. 2019;40(30):2534–47.
- 13 Atkinson G, Batterham AM, Thijssen DHJ, Green DJ. A new approach to improve the specificity of flow-mediated dilation for indicating endothelial function in cardiovascular research. J Hypertens. 2013;31(2):287–91.
- 14 Correale M, Leopizzi A, Mallardi A, Ranieri A, Suriano MP, D'Alessandro D, et al. Switch to direct anticoagulants and improved endothelial function in patients with chronic heart failure and atrial fibrillation. Thromb Res. 2020;195:16–20.

- 15 Celermajer DS, Sorensen KE, Gooch VM, Spiegelhalter DJ, Miller OI, Sullivan ID, et al. Non-invasive detection of endothelial dysfunction in children and adults at risk of atherosclerosis. Lancet. 1992;340(8828):1111–5.
- 16 Alaaeddine RA, AlZaim I, Hammoud SH, Arakji A, Eid AH, Abd-Elrahman KS, et al. The pleiotropic effects of antithrombotic drugs in the metabolic-cardiovascularneurodegenerative disease continuum: impact beyond reduced clotting. Clin Sci. 2021; 135(8):1015–51.
- 17 Ikari Y, Saito F, Kiyooka T, Nagaoka M, Kimura M, Furuki T, et al. Switching from Warfarin to rivaroxaban induces sufficiency of vitamin K and reduction of arterial stiffness in patients with atrial fibrillation. Heart Vessels. 2020;35(12):1727–33.
- 18 Namba S, Yamaoka-Tojo M, Kakizaki R, Nemoto T, Fujiyoshi K, Hashikata T, et al. Effects on bone metabolism markers and arterial stiffness by switching to rivaroxaban from warfarin in patients with atrial fibrillation. Heart Vessels. 2017;32(8):977–82.
- 19 Ageno W, Gallus AS, Wittkowsky A, Crowther M, Hylek EM, Palareti G. Oral anticoagulant therapy: antithrombotic therapy and prevention of thrombosis, 9th ed: American college of chest physicians evidence-based clinical practice guidelines. Chest. 2012;141(2 Suppl l):e44S–88S.
- 20 Giannitsi S, Bougiakli M, Bechlioulis A, Naka K. Endothelial dysfunction and heart failure: a review of the existing bibliography with emphasis on flow mediated dilation. JRSM Cardiovasc Dis. 2019;8:2048004019843047.
- 21 Heshmat-Ghahdarijani K, Jangjoo S, Amirpour A, Najafian J, Khosravi A, Heidarpour M, et al. Endothelial dysfunction in patients with lone atrial fibrillation. ARYA Atheroscler. 2020;16(6):278–83.
- 22 Lenart-Migdalska A, Drabik L, Kaźnica-Wiatr M, Tomkiewicz-Pająk L, Podolec P, Olszowska M. Increased levels of platelets and endothelial-derived microparticles in patients with non-valvular atrial fibrillation during rivaroxaban therapy. Clin Appl Thromb Hemost. 2021;27:10760296211019465.
- 23 Hwang CL, Piano MR, Phillips SA. The effects of alcohol consumption on flowmediated dilation in humans: a systematic review. Physiol Rep. 2021;9(10):e14872.
- 24 Oda N, Kajikawa M, Maruhashi T, Iwamoto Y, Kishimoto S, Matsui S, et al. Endothelial function is impaired in relation to alcohol intake even in the case of light alcohol consumption in Asian men; Flow-mediated Dilation Japan (FMD-J) Study. Int J Cardiol. 2017;230:523–8.
- 25 Voskoboinik A, Prabhu S, Ling LH, Kalman JM, Kistler PM. Alcohol and atrial fibrillation: a sobering review. J Am Coll Cardiol. 2016; 68(23):2567–76.
- 26 Shantsila A, Dwivedi G, Shantsila E, Butt M, Beevers DG, Lip GYH. Persistent macro-

vascular and microvascular dysfunction in patients with malignant hypertension. Hypertension. 2011;57(3):490-6.

- 27 Siasos G, Mazaris S, Zisimos K, Oikonomou E, Kokkou E, Konsola T, et al. The impact of atrial fibrillation on endothelial dysfunction. J Am Coll Cardiol. 2015;65(10):A477.
- 28 Komatsu T, Kunugita F, Ozawa M, Satoh Y, Yoshizawa R, Owada S, et al. Relationship between impairment of the vascular endothelial function and the CHA<sub>2</sub>DS<sub>2</sub>-VASc score in patients with sinus rhythm and non-valvular atrial fibrillation. Intern Med. 2018;57(15):2131–9.
- 29 Zhou J, Li YS, Chien S. Shear stress-initiated signaling and its regulation of endothelial function. Arterioscler Thromb Vasc Biol. 2014;34(10):2191–8.
- 30 Parker BA, Ridout SJ, Proctor DN. Age and flow-mediated dilation: a comparison of dilatory responsiveness in the brachial and popliteal arteries. Am J Physiol Heart Circ Physiol. 2006;291(6):H3043–9.
- 31 Frick M, Alber HF, Hügel H, Schwarzacher SP, Pachinger O, Weidinger F. Short- and long-term changes of flow-mediated vasodilation in patients under statin therapy. Clin Cardiol. 2002;25(6):291–4.
- 32 Schinzari F, Iantorno M, Campia U, Mores N, Rovella V, Tesauro M, et al. Vasodilator responses and endothelin-dependent vasoconstriction in metabolically healthy obesity and the metabolic syndrome. Am J Physiol Endocrinol Metab. 2015;309(9):E787–92.
- 33 Karelis AD, Faraj M, Bastard J-P, St-Pierre DH, Brochu M, Prud'homme D, et al. The metabolically healthy but obese individual presents a favorable inflammation profile. J Clin Endocrinol Metab. 2005;90(7):4145–50.
- 34 Wang L, Cheng CK, Yi M, Lui KO, Huang Y. Targeting endothelial dysfunction and inflammation. J Mol Cell Cardiol. 2022;168:58–67.
- 35 Iantorno M, Campia U, Di Daniele N, Nistico S, Forleo G, Cardillo C, et al. Obesity, inflammation and endothelial dysfunction. J Biol Regul Homeost Agents. 2014;28(2):169–76.
- 36 Christersson C, Wallentin L, Andersson U, Alexander JH, Alings M, De Caterina R, et al. Effect of apixaban compared with warfarin on coagulation markers in atrial fibrillation. Heart. 2019;105(3):235–42.
- 37 Broxterman RM, Witman MA, Trinity JD, Groot HJ, Rossman MJ, Park SY, et al. Strong relationship between vascular function in the coronary and brachial arteries. Hypertension. 2019;74(1):208–15.
- 38 Junejo RT, Braz ID, Lucas SJ, van Lieshout JJ, Phillips AA, Lip GY, et al. Neurovascular coupling and cerebral autoregulation in atrial fibrillation. J Cereb Blood Flow Metab. 2020; 40(8):1647–57.
- 39 Junejo RT, Braz ID, Lucas SJE, van Lieshout JJ, Lip GYH, Fisher JP. Impaired cerebrovascular reactivity in patients with atrial fibrillation. J Am Coll Cardiol. 2019;73(10):1230–2.